#### ORIGINAL ARTICLE



# Physiologic osteoclasts are not sufficient to induce skeletal pain in mice

Larissa de Clauser<sup>1,2</sup> | Sonia Santana-Varela<sup>1</sup> | John N Wood<sup>1</sup> | Shafaq Sikandar<sup>1,3</sup>

<sup>1</sup>Molecular Nociception Group, Wolfson Institute for Biomedical Research, University College London, London, UK

<sup>2</sup>Wolfson Centre for Age-Related Diseases, Institute of Psychiatry, Psychology and Neuroscience (IoPPN), King's College London, London, UK

<sup>3</sup>William Harvey Research Institute, Barts and the London School of Medicine & Dentistry, Mary University of London, London, UK

#### Correspondence

Larissa de Clauser, Wolfson Centre for Age-Related Diseases, Institute of Psychiatry, Psychology and Neuroscience (IoPPN), King's College London SE1 1UL London, UK.

 $Email: larissa.de\_clauser@kcl.ac.uk$ 

#### **Funding information**

This work was supported by the European Commission's Horizon 2020 Research and Innovation Programme (Marie Skłodowska-Curie grant 642720), the Wellcome Trust and Versus Arthritis.

#### Abstract

**Background:** Increased bone resorption is driven by augmented osteoclast activity in pathological states of the bone, including osteoporosis, fracture and metastatic bone cancer. Pain is a frequent co-morbidity in bone pathologies and adequate pain management is necessary for symptomatic relief. Bone cancer is associated with severe skeletal pain and dysregulated bone remodelling, while increased osteoclast activity and bone pain are also observed in osteoporosis and during fracture repair. However, the effects of altered osteoclast activity and bone resorption on nociceptive processing of bone afferents remain unclear.

**Methods:** This study investigates whether physiologic osteoclasts and resulting changes in bone resorption can induce skeletal pain. We first assessed correlation between changes in bone microarchitecture (through  $\mu$ CT) and skeletal pain using standardized behavioural phenotyping assays in a mouse model of metastatic bone cancer. We then investigated whether increased activity of physiologic osteoclasts, and the associated bone resorption, is sufficient to induce skeletal pain using mouse models of localized and widespread bone resorption following administration of exogenous receptor activator of nuclear factor kappa-B ligand (RANKL).

**Results:** Our data demonstrates that mice with bone cancer exhibit progressive pain behaviours that correlate with increased bone resorption at the tumour site. Systemic RANKL injections enhance osteoclast activity and associated bone resorption, without producing any changes in motor function or pain behaviours at both early and late timepoints.

**Conclusion:** These findings suggest that activation of homeostatic osteoclasts alone is not sufficient to induce skeletal pain in mice.

**Significance statement:** The role of osteoclasts in peripheral sensitization of sensory neurones is not fully understood. This study reports on the direct link between oestrogen-independent osteoclast activation and skeletal pain. Administration of exogenous receptor activator of nuclear factor kappa-B ligand (RANKL) increases bone resorption, but does not produce pro-nociceptive changes in behavioural pain thresholds. Our data demonstrates that physiologic osteoclasts are not essential for skeletal pain behaviours.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

© 2020 The Authors. European Journal of Pain published by John Wiley & Sons Ltd on behalf of European Pain Federation - EFIC®

Several pathological bone conditions, including metastatic bone cancer, osteoporosis and fracture repair, are characterized by increased bone resorption and structural changes in bone microarchitecture. Pain is frequently the biggest symptomatic burden in these bone pathologies, but the effect of altered osteoclast activity on nociceptive processing remains unclear.

Cancer-induced bone pain (CIBP) is a unique and complex condition that occurs as result of primary tumours metastasizing to the bone and involves mechanisms that are characteristic of inflammation, neuropathy, nerve compression and ischaemia (Falk & Dickenson, 2014; Honore & Mantyh, 2000). In animal models of CIBP osteoclast activity is increased (Currie et al., 2013). Pharmacological agents that target osteoclasts include bisphosphonates and denosumab, the latter an inhibitor of receptor activator of nuclear factor kappa-B ligand (RANKL), which is involved in osteoclast differentiation (Kong et al., 1999; Matsuzaki et al., 1998). Both bisphosphonates and denosumab are clinically licensed drugs that reduce bone loss, and in some cancer patients also relieve pain (Body et al., 2004; Lipton & Balakumaran, 2012).

In humans, osteoporotic pain is often only present after vertebral fractures and resolves after healing (Kim & Vaccaro, 2006), but the relationship between bone integrity and pain outcomes is inconsistent (Li et al., 2010). While in osteoarthritic patients, pain correlates with reduced tibial bone mineral density (BMD; Burnett et al., 2017), low back pain is associated with increased lumbar BMD (Lee et al., 2013). Models of osteoporosis involving ovariectomized rodents show mechanical hyperalgesia (Li et al., 2014; Sanoja & Cervero, 2005) and reduced BMD, which can both be reversed by targeting osteoclast activity with bisphosphonate therapy (Abe et al., 2015).

The initial, sharp pain produced by fractures in humans (Santy & Mackintosh, 2001) is generally reduced by repositioning and immobilization of the fractured bone (Camuso, 2006; Crandall et al., 2002). Similarly, in rodent fracture models a reduction in weight bearing on the affected limb is observed (Jimenez-Andrade et al., 2009). During fracture healing, osteoclasts show low-grade activity already during the early inflammatory phase of bone remodelling, which persists to increase for the duration of the healing process (Schell et al., 2006). Prolonged administration of bisphosphonates in rats impairs fracture healing (Savaridas et al., 2013), whereas a single bolus does not delay fracture repair (McDonald et al., 2008). These observations suggest that osteoclast activity is important throughout the healing process. The temporal role of osteoclast-mediated bone resorption during fracture repair, osteoporosis and bone cancer pain suggests that osteoclasts may play a pro-nociceptive role or mediate the resolution of pain in bone pathologies.

The aim of this study was to determine the contribution of osteoclasts to bone pain. We hypothesized that physiologic osteoclast activity alone is sufficient to induce skeletal pain. We first determined the extent of bone resorption and pain behaviours in a syngeneic model of CIBP. We then administered exogenous RANKL, locally within the femur and systemically, to induce bone resorption through physiologic osteoclast activation and assessed the effects on pain behaviour.

#### 2 | MATERIALS AND METHODS

#### 2.1 | Cell culture

LL/2 Lewis Lung carcinoma cells (ATCC) were cultured in DMEM supplemented with 10% of FBS and 1% of Penicillin/ Streptomycin for at least 2 weeks prior to surgery. Cells were split at 70%–80% confluence 1–2 days prior to surgery (cell culture reagents supplied by Thermo Fisher). On the day of surgery cells were harvested with 0.05% of Trypsin-EDTA, resuspended in DMEM at a final concentration of  $2 \times 10^7$  cells/ml and kept on ice till use. Cells were counted before and after intrafemoral injection to confirm viability.

#### 2.2 | Mice

A total of 88 C57Bl/6J mice (80 male and 8 female, 10–14 weeks old) purchased from Charles River and Envigo were used in this study. Female mice were used only for the systemic RANKL injection model to assess changes in BMD after 24 hr. Mice were housed in groups of 2–5 mice with a 12 hr light/dark cycle and allowed free access to water and standard diet. All mice were acclimatized for 2 weeks before the start of the experiment and randomized in experimental groups across cages. All experiments were performed with approval of personal and project licenses from the United Kingdom Home Office according to guidelines set by the Animals (Scientific Procedures) Act 1986 Amendment Regulations 2012, as well as guidelines of the Committee for Research and Ethical Issues of IASP.

#### 2.3 | Surgery

We used an established model of CIBP involving administration of cancer cells into the distal femoral marrow (Bangash et al., 2018; Huang et al., 2008; Isono et al., 2011; Minett et al., 2014; Yang et al., 2018). Briefly, mice were anaesthetized either with a mixture ketamine (125 mg/kg)/xylazine (13 mg/kg) (late stage cancer) or isoflurane (all other experiments) and sterile lacri-lube applied to their eyes. An incision was made in the skin above the patella. The patella and the lateral retinaculum tendons were loosened to move the patella to the side and expose the distal femoral epiphysis (Falk et al., 2013). A 30G needle was used to make a hole through the growth plate till the medullary cavity was reached. The needle was removed and a 30G syringe was used to introduce 10  $\mu$ l of either 2 × 10<sup>5</sup> carcinoma cells in DMEM medium, 10  $\mu$ g RANKL (human recombinant RANKL, Oriental Yeast Co., 47187000) in 1 mM of EDTA in PBS or vehicle. The hole was closed with bone wax (Johnson & Johnson) and the wound irrigated with sterile saline. The patella was moved back in place. The skin was sutured with 6–0 absorbable vicryl rapid (Ethicon) and lidocaine spray (Intubeaze, 20 mg/ml, Dechra) was applied to the superficial incision site.

#### 2.4 | RANKL administration

To reproduce the previously described widespread bone loss model (Tomimori et al., 2009), mice were administered 1 or 2 mg/kg RANKL (or 1 mM EDTA in PBS for the vehicle group) injected intraperitoneally daily for 2 days. As RANKL promotes the migration of osteoclast precursors towards bone (Kotani et al., 2013), we also used RANKL in a priming paradigm where mice were primed with an i.p. injection of 2 mg/kg RANKL, followed by a local intrafemoral injection of 10  $\mu$ l RANKL after 24 hr. Mice in the vehicle group were injected with 1 mM of EDTA in PBS.

#### 2.5 | Behavioural tests

For behavioural experiments, mice were acclimatized to the equipment for at least 2 days prior to testing. The experimenter was blind to the groups. For experiments involving intrafemoral injections, any mice with a limb score (LS) of less than 3 at 4 days after surgery were excluded to ensure pain outcomes were not due to the surgical procedure. Based on the criterion, we excluded the following mice: two animals for late stage cancer, one animal for intrafemoral RANKL injection, one animal for the priming i.p. followed by intrafemoral RANKL injection. Additionally, one animal from intermediate stage cancer was excluded as it failed to develop pain behaviour in the 20 day time period. This animal did not show a reduction in BMD.

#### 2.5.1 | Limb use score

Mice were allowed to freely move around in a glass box  $(30 \times 45 \text{ cm})$  for 5–10 min of acclimatization. Then, each mouse was observed for a period of 4 min and the use of the affected limb was scored as previously described (Falk

et al., 2013): 4 = Normal use of the limb; 3 = slight limping, characterized by preferential use of the unaffected limb when rearing; 2 = clear limping; 1 = clear limping and partial lack of use of the limb; 0 = lack of use of the affected limb during most of the observation time. LS = 2 was used as endpoint for intermediate and LS = 0 for late stage cancer.

#### 2.5.2 | Weight Bearing

Changes in weight bearing were assessed using an Incapacitance Metre (Linton Instrumentation) consisting of two scales. The mouse was allowed to place its head and upper body into a plastic tube to reduce stress and the hind limbs were positioned each on one of the scales. The load of each limb on the scale was measured for 5 s in which the mouse was still. Measurements were taken in triplicate, changing the position of the hind legs after each trial. The average weight-bearing ratio was calculated as the weight placed on the affected limb (left paw) divided by the total weight on both hind limbs:

$$\frac{\sum_{i=1}^{3} \left( \frac{\text{weight}_{\text{left}}}{\text{weight}_{\text{left}} + \text{weight}_{\text{right}}} \right)}{3} \times 100$$

#### 2.5.3 | Rotarod

The accelerating rotarod test is a behavioural paradigm to assess motor coordination (Jones & Roberts, 1968). Mice were placed on top of an accelerating rod (Life Science Series 8) and the rotation speed was set at 4 rpm. This speed was maintained for 30 s to allow mice to acclimatize. Time spent on the rotarod was recorded from the end of this period, at which point the speed was increased from 4 to 40 rpm in a 3 min ramp with a cut-off point of 5 min from the start of acceleration. The time was recorded till the animal fell from the rotarod. The test was repeated three times and the times averaged for each animal.

#### 2.5.4 | von Frey

Mechanical hypersensitivity was measured using the updown method to obtain the 50% withdrawal threshold (Chaplan et al., 1994). In brief, mice were placed in darkened enclosures with wire mesh floor and left to habituate for at least 1 hr till movement was reduced to a minimum. Filaments were applied perpendicular to the plantar surface for 3 s. Interval between stimuli was at least 30 s. Starting from a 0.4 g filament, the response was recorded as negative for no reaction or as positive for paw withdrawal. A positive response resulted in a decrease in filament strength for the



next stimulation, a negative response in increased strength. To determine the optimal threshold six responses in proximity of the 50% threshold are required. Thus, starting from the point at which the response to a filament changed from positive to negative or negative to positive, five further responses were recorded. When continuous positive responses were observed to the minimum stimulus of 0.008 g this set cut-off was used as the 50% withdrawal threshold. In the other cases, the pattern of responses obtained was used to calculate the 50% threshold =  $(10[\chi + \kappa \delta])/10,000)$ , where  $\chi$  is the log of the final von Frey filament used,  $\kappa$  the tabular value for the pattern of responses and  $\delta$  the mean difference between filaments (in log units).

### 2.5.5 | Hotplate

The mouse was placed into the hotplate apparatus (Ugo Basile), which was held at a temperature of  $50^{\circ}$ C. The test ended when the animal showed a withdrawal response or licked one of the hind paws. Cut-off time was set to 45 s.

#### 2.5.6 | Cold-plantar test

Cold-plantar test was used as previously reported (Brenner et al., 2012). Paw-withdrawal latency to the application of a pellet of compressed dry ice to the affected paw through a glass surface (6 mm thick) was recorded. Time for withdrawal was measured with a cut-off of 45 s. Testing was repeated for three times, with a waiting period of 15 min between stimulations.

#### 2.6 | Tissue processing

At the end of the experiment, mice were euthanized using carbon dioxide. For the CIBP model, animals were sacrificed paired with a sham when a LS of either 2 (intermediate stage) or 0 (late stage) was reached. Blood was collected through cardiac puncture, kept at RT for 30 min to 1 hr, and then, centrifuged for 10 min at 2000 g. Supernatant serum was collected and kept in  $-80^{\circ}$ C till further use. Femurs were collected and fixed in 4% of PFA for 24 hr. They were then transferred to 70% of EtOH until micro-CT scanning.

### 2.6.1 | Quantitative analysis of TRAP in serum

TRAP solution buffer was obtained by adding freshly made 50 mM of L-Ascorbic acid, 20 mM of disodium tartrate and 80 mM of 4-nitrophenylphosphate to TRAP reaction buffer

(1 M Acetate, 0.5% Triton X-100, 1 M NaCl, 10 mM EDTA, pH = 5.5) and ultrapure water at a ratio of 1:1:1:2:3. Serum samples were diluted 1:10 and 20  $\mu$ l of each sample were loaded in triplicate into a 96 flat bottom well plate (Nunc-Immuno). 80  $\mu$ l of freshly made TRAP solution buffer was added to each well. The microplate was incubated at 37°C for 1 hr in the dark. The reaction was stopped by adding 100  $\mu$ l of 0.3 M NaOH to each well. Absorbance was measured at 405 nm using a microplate reader as previously described (Karsdal et al., 2005).

#### 2.6.2 | Micro-computed tomography (µCT)

Images were acquired with SkyScan software at 6.41 µm/ pixel with 0.6° rotation steps and 2 frame averaging. Image data were reconstructed with NRecon software. CT-analyser was used to select a 1 mm volume of interest (VOI) region, starting at 0.6 mm from the growth plate of the distal femur. Separate VOIs where created for cortical (CIBP) or trabecular (all models) bone surface through manual selection. BMD was quantified by plotting attenuation coefficients against a standard curve determined with two phantoms with known density (250 and 750 mg/cm<sup>3</sup>). BoneJ2, a collection of skeletal biology plug-ins for ImageJ, was used for microarchitecture analysis (Doube et al., 2010). The 1 mm VOI was binarized and standardized 3D geometric parameters were determined, including total volume (TV), total bone volume (BV), bone surface (BS), trabecular thickness (Tb.Th.) and trabecular spacing (Tb.Sp.). To determine trabecular number (Tb.N.) and connectivity density (~number of trabeculae per unit volume), the images were first purified using a chunk size of four slices and a mapped labelling algorithm, to obtain a single connected structure throughout the image stack.

#### 2.7 | Statistics

Using G\*Power (Faul et al., 2007), we calculated sample size using two-tailed *t*-test, on alpha level = 0.05, power = 0.80 and effect size = 2.8406 (based on BMD from late stage cancer). We found that the minimum needed sample size was four mice per group. Statistical analysis was performed with GraphPad Prism 8. Kruskal–Wallis test was used to determine if LS at the endpoint significantly differed between groups. Unpaired Welch's *t*-test (two-tailed) was used to compare difference between two distributions. Comparison of multiple timepoints between groups was performed by two-way ANOVA with Bonferroni post hoc test. For data with missing timepoints mixed effects model REML with Bonferroni post hoc test was used instead. Data are presented as mean  $\pm$  standard error of the mean (SEM) and significance as: \*p < .05; \*\*p < .01; \*\*\*p < .001, \*\*\*\*p < .0001.

#### 3 | RESULTS

## **3.1** | CIBP is associated with a reduction in limb use and weight bearing

We used a previously described syngeneic model of metastatic bone cancer to investigate the correlation between bone degradation and pain behaviour (Bangash et al., 2018; Huang et al., 2008; Isono et al., 2011; Minett et al., 2014; Yang et al., 2018). We used the LS of the affected hindpaw to quantify the pain phenotype associated with metastatic bone cancer. In two independent studies, we defined the endpoint either as 'intermediate stage' when the mice showed clear signs of limping (LS = 2) or 'late stage' when the use of the affected limb was completely avoided (LS = 0). At the endpoint LS significantly differed between groups (Kruskal–Wallis test: \*\*\*\*p < .0001). Corresponding survival curves show that mice reached a limb use score of 2 between days 8 and 15, whereas mice reached a LS of 0 at a later timepoint, between days 12 and 15 (Figure 1a). All cancer mice showed a significantly reduced survival compared to sham mice (Log-rank (Mantel-Cox) test: \*\*\*p = .0001). Additionally, weight bearing on the affected limb was assessed. Both intermediate and late stage cancer mice showed a significant reduction in weight bearing on the affected limb compared to sham mice when they reached their respective endpoints of the study (Figure 1b, two-way ANOVA with Bonferroni post hoc: \*\*\*\*p < .0001). This effect was more pronounced in the late stage cancer mice (baseline versus endpoint; two-way ANOVA with Bonferroni post hoc: \*\*\*p = .0001) and is highlighted by a significant linear correlation between limb use score and percentage weight bearing on the affected limb (Figure 1c, Pearson correlation: \*\*\*\*p < .0001, interpolated linear curve: y = 4.745x + 29.910,  $r^2 = .7595$ ).

### 3.1.1 | Intermediate and late stage cancer are associated with bone resorption

We measured serum TRAP as a marker for increased osteoclast activity, but found no differences between late stage cancer and sham mice (Figure 1d). However, trabecular BMD was reduced by ~37% in intermediate cancer stage mice ( $202 \pm 18.64$  vs.  $321.1 \pm 14.39$  mg/cm<sup>3</sup>; Welch's *t*test: \*\*\*p = .0003) and by ~25% in late stage cancer mice ( $202 \pm 18.64$  vs.  $321.1 \pm 14.39$  mg/cm<sup>3</sup>; Welch's *t*-test: \*\*\*p = .0010) compared to respective sham mice (Figure 1e). Cortical BMD was reduced by ~16% in intermediate stage cancer mice ( $555.8 \pm 26.96$  vs.  $665.4 \pm 13.17$  mg/cm<sup>3</sup>; Welch's *t*-test: \*\*p = .0056) and by ~17% in late stage cancer mice ( $562.7 \pm 28.83$  vs.  $682.9 \pm 20.61$  mg/cm<sup>3</sup>; Welch's *t*-test: \*\*p = .0042), compared to respective sham mice P

5

(Figure 1f). Representative  $\mu$ CT images and 3D reconstruction of the analysed 1 mm volume are shown in Figure 1g. We found a significant positive correlation between BMD and skeletal pain, the latter measured as percentage weight bearing on the affected limb (Figure 1h, Pearson correlation: \*p = .0468, interpolated linear curve: y = 0.4278x + 1.759,  $r^2 = .1547$ ). Sustained reductions in trabecular and cortical BMD indicate ongoing osteoclast activity that augments bone resorption throughout intermediate to late stage cancer in the mouse model.

## **3.2** | A localized injection of RANKL is not sufficient to produce bone resorption and pain behaviour

To investigate the role of local physiologic osteoclasts in skeletal pain, we first activated locally resident osteoclasts in the mouse femur using 10 µl of RANKL (1 mg/ml). A single intrafemoral administration of RANKL did not produce significant changes in pain behaviour or BMD (Fig. S1). Therefore, we resorted to priming mice with a systemic injection of 2 mg/kg i.p. RANKL (or vehicle) in order to promote the migration of osteoclast precursors towards bone (Kotani et al., 2013), followed by a local intrafemoral injection of 10 µl RANKL (or vehicle) on the following day (Figure 2a). Both vehicle- and RANKL-injected mice showed normal motor function, assessed as time spent on rotarod (Figure 2b, two-way ANOVA with Bonferroni post hoc, p = .9608). Mice did not show any changes in weight bearing on the affected limb (Figure 2c, two-way ANOVA with Bonferroni post hoc, p = .4935), and did not show mechanical hypersensitivity (Figure 2d, two-way ANOVA with Bonferroni post hoc, p = .6040). Similar responses to noxious heat stimulation (Figure 2e, two-way ANOVA with Bonferroni post hoc, p = .3571) and noxious cold stimulation (Figure 2f, two-way ANOVA with Bonferroni post hoc, p = .8013) were observed between RANKL- and vehicle-injected mice. Serum levels of the bone resorption biomarker TRAP (Figure 2g, Welch's *t*-test: p = .0973) and trabecular BMD (Figure 2h, Welch's *t*-test: p = .8499) did not differ between RANKL- and vehicle-injected mice. The results together indicate that the combination of systemic and localized RANKL administration does not produce structural changes in bone integrity or pain behaviour.

## **3.3** | A model of widespread bone resorption produces sex-specific and temporal changes in BMD

To determine whether the absence of bone resorption in Figure 2 was related to a low dose of RANKL used





**FIGURE 1** Pain behaviour and structural outcomes in a mouse model of cancer-induced bone pain. (a) Survival curve after surgery for sham (dark grey line, n = 14), intermediate (blue line, n = 7) and late stage cancer (red line, n = 5) mice with endpoint defined as limping (LS = 2) and non-use of the affected limb (LS = 0), respectively. (b) Weight bearing on the affected limb at baseline and endpoint in sham mice (black bar for intermediate stage, n = 7; light grey for late stage, n = 7), intermediate stage cancer mice (blue, n = 7) and late stage cancer mice (red, n = 5). (c) Correlation between limb use score and weight bearing, showing the median (straight line), the 25–75th percentile (boxes), the 5–95th percentile (whiskers) and interpolated linear curve with 95% confidence interval (dotted line, grey area). (d) Systemic levels of TRAP in serum of sham and late stage cancer mice. (e) Trabecular BMD and (f) cortical BMD in intermediate and late stage cancer mice. (g) Representative  $\mu$ CT images from sham, intermediate stage cancer and later stage cancer femurs. (h) Correlation between log transformed trabecular BMD and weight bearing on the affected limb, showing single values with interpolated linear curve (straight line) and 95% confidence interval (dotted lines). Data are shown as mean  $\pm$  SEM. \*\* $p \le .01$ , \*\*\* $p \le .001$ , \*\*\*\* $p \le .001$  for differences between groups

in the intrafemoral injections, we used a published murine model of widespread induction of bone resorption (Tomimori et al., 2009) that involves consecutive i.p. injections of 1 mg/kg of RANKL (Figure 3a). Fortyeight hour after RANKL injections, we found a significant reduction in trabecular BMD in male (Welch's *t*-test \*p = .0443), but not female mice (Figure 3b; Welch's *t*-test, p = .2675). As a bone resorptive phenotype was observed at 48 hr, a higher dose of 2 mg/kg RANKL was administered on two consecutive days in male mice via i.p. injection and pain behaviour was closely followed over a period of 20 days (Figure 4a). Motor behaviour was not impaired in RANKL-injected mice compared to vehicle, as assessed through time spent on rotarod (Figure 4b, two-way ANOVA with Bonferroni post hoc, p = .3365). Mechanical sensory function was not significantly



**FIGURE 2** Effects of a combination of systemic and localized administration of RANKL on bone resorption and pain behaviour. (a) Schematic overview of the study design. Following a priming i.p. injection (black circle) mice received an intrafemoral injection (green circle) of RANKL or vehicle and their behaviour was closely followed for 20 days. (b) RANKL- (green, n = 4) and vehicle-injected mice (black, n = 6) were assessed for motor function measured through latency to fall on a rotarod. (c) Ongoing pain was measured through weight bearing on the injected limb. (d) Mechanical sensitivity was measured by withdrawal thresholds to von Frey filaments. Thermal sensitivity was measured as nociceptive latencies to (e) noxious heat and (f) noxious cold. (g) Serum levels for the osteoclast biomarker TRAP. (i) Trabecular BMD in the ipsilateral femur. Data are shown as mean  $\pm$  SEM

different between RANKL- and vehicle-injected mice (Figure 4c, mixed effects model REML with Bonferroni post hoc, p = .2277). Thermal pain thresholds on the hotplate (Figure 4d, two-way ANOVA with Bonferroni post hoc, p = .2706) and cold-plantar test (Figure 4e, two-way ANOVA with Bonferroni post hoc, p = .8533) were similar between groups. Moreover, 20 days after two consecutive i.p. injections of RANKL, systemic levels of the biomarker TRAP in serum did not differ between RANKL- and vehicle-injected mice (Figure 4f). In contrast to the observed loss of trabecular BMD at 48 hr using two RANKL i.p. administrations at a smaller dose (Figure 3), no discernible differences in overall trabecular BMD could be observed after 20 days using a higher dosing regime (Figure 4g, Welch's *t*-test: p = .1328). However, 3D reconstruction of µCT data suggests that a more localized reduction in BMD was present in most mice (Figure 4h). Further stratification of the data, by dividing the analysed 1 mm volumetric region in smaller segments of 0.2 mm, revealed a significant reduction in BMD in the two segments which were further away from the growth plate (Figure 4i, Welch's *t*-test: \*p = .0183 for 1.2-1.4 mm and \*\*p = .0057 for 1.4-1.6 mm, p > .1000for the other segments).

#### 3.4 Changes in bone microarchitecture is correlated with pain behaviour in a murine model of bone cancer, but not following induction of localized or widespread bone resorption

We quantified and compared changes of trabecular bone microarchitecture in cancer mice, RANKL-injected mice and their sham/vehicle counterparts (Table 1). In mice with intermediate stage cancer, several parameters including percentage bone volume fraction (%BV/TV) (Welch's t-test, \*\*\*p = .0006), bone surface fraction (BS/TV) (Welch's ttest, \*\*\*p = .0003), trabecular spacing (Tb.Sp.) (Welch's *t*test, \*p = .0416), trabecular number (Tb.N) (Welch's *t*-test, \*p = .0189) and connectivity density (Conn.D.) (Welch's *t*-test, \*p = .0139) were substantially reduced compared to sham mice. On the contrary, in late stage cancer mice, only a trend in changes of these parameters could be found. Twenty days after two consecutive systemic injections of 2 mg/kg RANKL changes in bone microarchitecture were still visible, with RANKL-injected mice showing significantly reduced BS/TV (Welch's *t*-test, \*\*\*p = .0006), Tb.N. (Welch's *t*-test, \*\*\*p = .0003), Conn.D. (Welch's *t*-test, \*\*p = .0013), whereas trabecular thickness (Tb.Th.) was substantially



**FIGURE 3** Effects of consecutive systemic administration of RANKL on bone resorption. (a) Schematic representation of the study design. Mice received two consecutive systemic injections of 1 mg/kg RANKL and bones were collected 2 days after the last injection. (b) Trabecular BMD in male and female mice injected with RANKL (yellow, n = 4 for female and n = 4 for male) or vehicle (black, n = 4 for female and n = 4 for male). (c) Representative  $\mu$ CT images from male and female vehicle- and RANKL-injected femurs. Data are shown as mean  $\pm$  SEM.  $*p \le .05$  for differences between groups

(c)





**FIGURE 4** Effects of consecutive, high-dose systemic administration of RANKL on pain behaviour and bone resorption. (a) Schematic representation of the study design. Mice received two consecutive systemic injections (black arrows) of 2 mg/kg RANKL (yellow, n = 6) or vehicle (black, n = 6). Behaviour was closely followed for 20 days, when bones were collected. (b) Mice were assessed for motor function, (c) mechanical sensitivity to von Frey filaments, (d) heat sensitivity and (e) cold sensitivity. (f) Serum levels for biomarker TRAP at 20 days. (g) Trabecular BMD of a 1 mm volumetric region starting at 0.6 mm from the growth plate. (h) Trabecular BMD for 0.2 mm thick volumetric sections at increasing distance from the growth plate. (i) Representative  $\mu$ CT images from vehicle- and RANKL-injected femurs. Data are shown as mean  $\pm$  SEM. \* $p \le .05$ , \*\* $p \le .01$  for differences between groups

	Intermediate stage		Late stage		20 days (male)		48 hr (male)		48 hr (female)	
Parameter	Sham	Cancer	Sham	Cancer	Vehicle	RANKL	Vehicle	RANKL	Vehicle	RANKL
Cross-section										$\bigcirc$
1 mm VOI			all and the	a start			Star Star	el a la contra		
BMD	$316.904 \pm 11.986$	$205.004 \pm 15.760^{***}$	$368.758 \pm 11.612$	$274.924 \pm 13.148^{***}$	$240.268 \pm 9.890$	$214.152 \pm 10.686$	$360.373 \pm 9.322$	$289.418 \pm 19.704^*$	$202.738 \pm 11.091$	$184.015 \pm 6.73$
%BV/TV	$9.369 \pm 0.644$	$4.656 \pm 0.694^{***}$	$10.007 \pm 0.561$	$8.010 \pm 1.021$	$10.061 \pm 0.509$	$8.620 \pm 0.484$	$10.484 \pm 0.639$	$7.395 \pm 0.768^{*}$	$4.549 \pm 0.270$	$3.359 \pm 0.145*$
BS/TV (mm <sup>-1</sup> )	$6.267 \pm 0.375$	$3.258 \pm 0.413^{***}$	$6.568 \pm 0.397$	$5.593 \pm 0.536$	$6.787 \pm 0.230$	$5.078 \pm 0.222^{***}$	$6.331 \pm 0.433$	$4.918 \pm 0.311$	$3.759 \pm 0.191$	$2.898 \pm 0.086^{*}$
Tb.Th. (mm)	$0.052 \pm 0.001$	$0.049 \pm 0.001$	$0.051 \pm 0.001$	$0.048 \pm 0.002$	$0.051 \pm 0.001$	$0.062 \pm 0.004^{*}$	$0.058 \pm 0.001$	$0.053 \pm 0.003$	$0.042 \pm 0.001$	$0.040 \pm 0.001$

Parameters of bone microarchitecture in C57BL/6J receiving different agents that promote osteoclast activity **TABLE 1** 

\*\*  $p \leq .01$ , \* $p \leq .05$ ). Representative pictures were chosen based on the average measured BMD and show binarized cross-section at 1 mm from the growth plate (upper row, scalebar 500 µm) and 3D reconstruction of the 1 mm Note: Data are shown as mean  $\pm$  standard error of the mean. For each treatment, data were compared with the respective control group and differences in distribution were determined by unpaired Welch's *t*-test (\*\*\* $p \leq .001$ , VOI (0.6-1.6 mm from growth plate) used to determine parameters for bone microarchitecture: % BV/TV (%bone volume fraction), BS/TV (bone surface fraction), Tb.Th. (trabecular thickness), Tb.Sp. (trabecular spacing), Tb.N. (number of trabeculae), Conn.D. (Connectivity density or number of trabeculae per volume).

 $140.875 \pm 11.636^{**}$  $47.674 \pm 4.333^{**}$ 

 $267.625 \pm 19.513$  $87.9670 \pm 6.740$ 

 $399.750 \pm 41.469$ 

 $469.125 \pm 46.396$  $0.553 \pm 0.015$ 

> $336.083 \pm 32.111^{***}$  $91.194 \pm 9.163^{**}$

 $589.500 \pm 14.431$  $151.946 \pm 6.373$ 

 $480.300 \pm 130.107$  $130.264 \pm 34.817$ 

 $207.071 \pm 42.560 *$  $0.715\pm 0.036^{**}$ 

> $909.643 \pm 205.335$  $217.472 \pm 44.269$

Tb.N. (mm<sup>-1</sup>) Tb.Sp. (mm)

 $(mm^{-3})$ Conn.D.

 $0.612\pm0.017$ 

 $54.831 \pm 10.766^*$ 

 $171.904 \pm 20.176$  $634.429 \pm 69.279$ 

 $0.591\pm0.012$ 

 $0.564\pm0.011$ 

 $0.550 \pm 0.019$ 

 $0.550 \pm 0.014$ 

 $128.926 \pm 11.003$   $111.227 \pm 8.762$ 

 $0.542\pm0.022$ 

 $0.544\pm0.017$ 

 $0.582 \pm 0.006$ 



increased (Welch's *t*-test, \*p = .0375) compared to vehicleinjected mice. While not significant, similar changes were present 48 hr after two consecutive injections of 1 mg/kg RANKL, with male mice showing a substantial decrease in BV/TV (Welch's *t*-test, \*\*p = .0013). We also found a substantial reduction in % BV/TV (Welch's *t*-test, \*\*p = .0227), BS/TV (Welch's *t*-test, \*\*p = .0220), Tb.N. (Welch's *t*-test, \*\*p = .0050) and Conn.D. (Welch's *t*-test, \*\*p = .0069) in female mice.

#### 4 | DISCUSSION

## 4.1 | Bone cancer is associated with a dysregulation in normal bone remodelling and correlates with pain behaviour

We used a syngeneic model of CIBP to determine outcomes on bone resorption and the development of chronic pain. Previous studies of CIBP have used mice at 6–8 weeks (Falk et al., 2013; Guedon et al., 2016; Minett et al., 2014) when sexual maturity is reached, but mice of this age are still undergoing extensive bone development. We used 12-week-old mice which have mature bone, defined by stable mechanical properties, bone mass and length (Ferguson et al., 2003). Moreover, at this stage the rate of bone mineralization and resorption reach a plateau. To our knowledge, this is the first study investigating the effects of CIBP in mice with fully developed bones.

In line with previous studies, we found a reduction in limb use and weight bearing on the affected limb in mice with bone cancer (Minett et al., 2014). There was a significant correlation between the two measures with the behavioural phenotypes deteriorating from intermediate stage cancer (defined by clear limping) to late stage cancer (where mice completely avoid use of the affected limb). Noteably, the extent of osteolytic lesions in both cortical and trabecular bone of the affected femur was more pronounced in intermediate stage cancer compared to late stage cancer mice. In accordance with a previous study (Falk et al., 2013), we found a positive correlation between bone degradation and skeletal pain. Our findings indicate a homeostatic equilibrium in bone degradation and/or new bone formation is already established when mice with bone cancer begin to reduce limb use (lower LSs that reflect limping), but does not progress further.

### 4.2 | The role of osteoclasts in a model of RANKL-induced bone resorption

RANKL can promote both osteoclastogenesis and recruitment of hematopoietic precursors from the bloodstream (Kotani et al., 2013). However, previous studies indicate that a single systemic injection of RANKL in mice is not sufficient to induce bone resorption (Tomimori et al., 2009). We therefore used the approach of intrafemoral injection of RANKL to induce activation of locally resident osteoclasts, both with and without a prior priming injection of systemic RANKL, but failed to observe evidence of dysregulated bone remodelling or pain behaviours. One possible explanation may be a dose-limiting factor of the injected concentration of 10 µg RANKL, which may not be enough to induce sustained osteoclast activation. However, due to the instability of RANKL at higher concentrations and the maximum volume capacity of the mouse femur (Zilber et al., 2008), we were not able to administer a higher dose. Instead, we used a model of widespread bone resorption induced by systemic RANKL injections (Tomimori et al., 2009) to determine whether physiologic osteoclasts can induce pain behaviour in mice. Surprisingly, we found a significant decrease in BMD in male, but not female mice, 48 hr after two consecutive systemic RANKL injections. Previous work shows osteoclast activation in both male and female mice (Ariza et al., 2019; Ding et al., 2020; Li et al., 2015; Shin et al., 2014; Tomimori et al., 2009). One possibility for the discrepancy between our data and other studies may be related to a reduction in the accuracy of trabecular BMDs below 250 mg/cm<sup>3</sup> when using two phantoms of calcium-hydroxyapatite of 250 and 750 mg/ cm<sup>3</sup> to estimate density. 3D reconstructions and detailed analysis of bone microarchitecture, incuding bone volume fraction, bone surface fraction, trabecular number and connectivity density suggest some degree of bone resorption is present in female mice.

Even at higher systemic doses of RANKL, the significant decrease in BMD observed after 48 hr in male mice was not sustained throughout the experimental period of 20 days. These data reflects previous descriptions of recovery of normal bone turnover over a prolonged time period (Tomimori et al., 2009). However, using 3D reconstructions, we observed a trend of localized effects on bone resorption. When dividing the volumetric region in smaller segments of 0.2 mm thickness, we found significantly reduced BMD in the two segments located further away from the growth plate, which is likely due to differentiation of chondrocytes into osteoblasts near the growth plate, resulting in faster bone formation (Mizuhashi et al., 2019; Newton et al., 2019).

Bone strength is influenced by several factors, including material properties, geometry and notably tissue microarchitecture (Dalle Carbonare & Giannini, 2004; Felsenberg & Boonen, 2005). In humans, trabecular microarchitecture parameters correlate with fracture risk independently of BMD (Elizabeth et al., 2019; Kijowski et al., 2012). Similarly, in trabecular samples of the femoral head, bone microarchitecture parameters are better predictors of bone strength and structure than BMD (Topoliński et al., 2012). In this study, alongside reduction in BMD, we also observed significantly reduced bone surface fraction, trabecular number and connectivity density in mice following systemic RANKL administration that indicates osteoclast-mediated bone resorption. Unexpectedly, trabecular thickness was significantly higher in RANKL-treated mice, potentially reflecting a reduction in small trabeculae in these mice. Similar findings have been reported in an OVX rat model of bone resorption: in young rats oestrogen deprivation leads to reduced trabecular thickness but this effect is reversed with ageing (Butler et al., 2007). Similarly, in C57BL/6J mice ovariectomy produces a reduction or an increase in trabecular thickness for 8 week old, and 12 weeks or older animals, respectively (Zhou et al., 2018).

In mice with CIBP, we observed changes in both trabecular and cortical bone, reflecting tumour-induced lesions (Bloom et al., 2011; Mantyh et al., 2010). Animals injected with RANKL, on the contrary, did not present any visible changes in cortical BMD. Therefore, the site of osteoclast activation may be important in determining pain development, as cortical lesions may sensitize periosteal afferents. However, sensitization of bone marrow afferents is also able to produce pain, as is observed in patients undergoing bone marrow aspiration (Gendron et al., 2019). Additionally, in osteoarthritic patients pain outcomes correlate with trabecular, but not cortical BMD (Burnett et al., 2017).

Overall, several trabecular bone microarchitecture parameters suggest significant alterations in bone microstructure in mice receiving two consecutive systemic injections of RANKL. However, the extent of osteoclast activation was not sufficient to induce pain behaviour, even at early timepoints when differences in BMD were more pronounced between RANKL- and vehicle-injected mice. Compared to other inbred mouse strains, C57BL/6J have only few trabeculae and less mineralized bone (Zhou et al., 2018). Other mouse strains may therefore constitute a better model to study the effects of osteoclast activation and bone resorption.

## 4.3 | Physiologic osteoclasts are not sufficient to induce skeletal pain

Although mice with late stage cancer and RANKL-treated mice show similar reductions in BMD and bone microarchitecture parameters, pain behaviour was present only in mice with CIBP. The observed skeletal pain phenotype in late stage cancer mice may depend on a phenotypic switch in osteoclasts present in pathological conditions. Metastatic lung cancer to the bone produces ostolytic lesions by altering the bone microenvironment to promote the vicious cyle of bone resorption and tumour growth (Mundy, 2002). Some factors released in the bone microenvironemnt by cancer and stromal cells, including interleukin-6 (IL-6) and tumour necrosis factor  $\alpha$  (TNF- $\alpha$ ) (Kim et al., 2009), are able to induce osteoclast differentiation in vitro independently of RANK activation (O'Brien et al., 2016). Blockade of IL-6 in a rat model of bone cancer reduced mechanical hypersensitivity and ongoing pain, and prevented bone degradation, without affecting tumour burden (Remeniuk et al., 2018). Evidence from skeletal inflammatory conditions suggests that IL-6 and TNF-a induce 'inflammatory osteoclasts', with a distinct phenotypic profile that can be compared to different macrophage subtypes (Novack, 2016). In ovariectomized mice, femur marrows show increased levels of the cytokines TNF- $\alpha$ , IL-6, and interleukin-1  $\beta$  (IL-1 $\beta$ ) (Kanaya et al., 2016). Similarly, in a model of injury to the intervertebral disc, anti-RANKL treatment reduces expression of TNF-a and IL-6 in the bone, while also preventing the increase of calcitonin gene-related peptide (CGRP) in retrogradely labelled DRG neurones (Sato et al., 2015). In a serumtransfer murine model of arthritis, pro-inflammatory cytkoines including TNF- $\alpha$  and IL-6 are present in the joint (Ji et al., 2002). Additionally, both RANK knockout and wild-type mice have osteoclasts in the arthritic synovium, although these cells are absent in the bone marrow compartment of knockout mice (O'Brien et al., 2016), suggesting a different origin from canonical osteoclastogeneis. It remains to be investigated if these so-called 'inflammatory osteoclasts' consitute a distinct population from homeostatic osteoclasts in terms of gene expression and resorptive properties, but our findings support the notion that increased activity of physiologic ostecoclasts does not contribute to peripheral sensitization in bone pathologies.

#### 5 | CONCLUSION

In summary, this study reveals that physiologic osteoclasts alone are unable to induce skeletal pain in mice. Several painful skeletal conditions, including bone cancer, are associated with increased osteoclast activity, as well as neuronal sprouting, suggesting that several mechanisms may act in synchrony to produce and maintain chronic bone pain. One such mechanism may involve the phenotypic switch towards an 'inflammatory osteoclasts' mediated by TNF- $\alpha$  and IL-6.

#### ACKNOWLEDGEMENTS

We thank Prof. Tim Arnett (University College London) for the use of the  $\mu$ CT scanner, advice and support. and Prof. Anne-Marie Heegaard (Denmark University) for advice and support.

#### **CONFLICT OF INTEREST**

The authors report no conflict of interest.

AUTHOR CONTRIBUTIONS

LdC, JW and SS conceived the study, designed the experiments and interpreted the results. LdC, SS and SSV performed the experiments. LdC analysed the data. LdC, SS and JW prepared the manuscript. All contributing authors discussed the results, commented on the manuscript and agreed to submission for publication.

#### REFERENCES

- Abe, Y., Iba, K., Sasaki, K., Chiba, H., Kanaya, K., Kawamata, T., Oda, K., Amizuka, N., Sasaki, M., & Yamashita, T. (2015). Inhibitory effect of bisphosphonate on osteoclast function contributes to improved skeletal pain in ovariectomized mice. *Journal of Bone and Mineral Metabolism*, 33(2), 125–134. https://doi.org/10.1007/s0077 4-014-0574-x
- Ariza, Y., Murata, M., Ueda, Y., & Yoshizawa, T. (2019). Bruton's tyrosine kinase (Btk) inhibitor tirabrutinib suppresses osteoclastic bone resorption. *Bone Reports*, 10, 100201. https://doi.org/10.1016/j. bonr.2019.100201
- Bangash, M. A., Alles, S. R. A., Santana-Varela, S., Millet, Q., Sikandar, S., de Clauser, L., ter Heegde, F., Habib, A. M., Pereira, V., Sexton, J. E., Emery, E. C., Li, S., Luiz, A. P., Erdos, J., Gossage, S. J., Zhao, J., Cox, J. J., & Wood, J. N. (2018). Distinct transcriptional responses of mouse sensory neurons in models of human chronic pain conditions. *Wellcome Open Research*, *3*, 78. https://doi. org/10.12688/wellcomeopenres.14641.1. https://doi.org/10.12688/ wellcomeopenres.14641.1
- Bloom, A. P., Jimenez-Andrade, J. M., Taylor, R. N., Castañeda-Corral, G., Kaczmarska, M. J., Freeman, K. T., Coughlin, K. A., Ghilardi, J. R., Kuskowski, M. A., & Mantyh, P. W. (2011). Breast cancer-induced bone remodeling, skeletal pain, and sprouting of sensory nerve fibers. *Journal of Pain*, 12(6), 698–711. https://doi. org/10.1016/j.jpain.2010.12.016
- Body, J. J., Diel, I. J., Bell, R., Pecherstorfer, M., Lichinitser, M. R., Lazarev, A. F., Tripathy, D., & Bergström, B. (2004). Oral ibandronate improves bone pain and preserves quality of life in patients with skeletal metastases due to breast cancer. *Pain*, 111(3), 306–312. https://doi.org/10.1016/j.pain.2004.07.011
- Brenner, D. S., Golden, J. P., & Gereau, R. W., IV (2012). A novel behavioral assay for measuring cold sensation in mice. *PLoS One*, 7(6), e39765. https://doi.org/10.1371/journal.pone.0039765
- Burnett, W. D., Kontulainen, S. A., McLennan, C. E., Hazel, D., Talmo, C., Wilson, D. R., Hunter, D. J., & Johnston, J. D. (2017). Proximal tibial trabecular bone mineral density is related to pain in patients with osteoarthritis. *Arthritis Research and Therapy*, 19(1), 1–9. https://doi.org/10.1186/s13075-017-1415-9
- Butler, R. J., Marchesi, S., Royer, T., & Davis, I. S. (2007). The Effect of a subject-specific amount of lateral wedge on knee. *Journal* of Orthopaedic Research, 25(May), 1121–1127. doi:https://doi. org/10.1002/jor.
- Camuso, M. R. (2006). Far-forward fracture stabilization: External fixation versus splinting. *Journal of the American Academy* of Orthopaedic Surgeons, 14(10), S118–S123. https://doi. org/10.5435/00124635-200600001-00027
- Chaplan, S. R., Bach, F. W., Pogrel, J. W., Chung, J. M., & Yaksh, T. L. (1994). Quantitative assessment of tactile allodynia in the rat paw. *Journal of Neuroscience Methods*, 53(1), 55–63. https://doi. org/10.1016/0165-0270(94)90144-9

- Crandall, M., Miaskowski, C., Kools, S., & Savedra, M. (2002). The pain experience of adolescents after acute blunt traumatic injury. *Pain Management Nursing*, 3(3), 104–114. https://doi.org/10.1053/ jpmn.2002.126070
- Currie, G. L., Delaney, A., Bennett, M. I., Dickenson, A. H., Egan, K. J., Vesterinen, H. M., Sena, E. S., Macleod, M. R., Colvin, L. A., & Fallon, M. T. (2013). Animal models of bone cancer pain: Systematic review and meta-analyses. *Pain*, 154(6), 917–926. https://doi.org/10.1016/j.pain.2013.02.033
- Dalle Carbonare, L., & Giannini, S. (2004). Bone microarchitecture as an important determinant of bone strength. *Journal* of Endocrinological Investigation, 27(1), 99–105. https://doi. org/10.1007/bf03350919
- Ding, N., Lu, Y., Cui, H., Ma, Q., Qiu, D., Wei, X., Dou, C., & Cao, N. (2020). Physalin D inhibits RANKL-induced osteoclastogenesis and bone loss via regulating calcium signaling. *BMB Reports*, 53(3), 154–159. https://doi.org/10.5483/bmbrep.2020.53.3.147
- Doube, M., Klosowski, M. M., Arganda-Carreras, I., Cordelières, F. P., Dougherty, R. P., Jackson, J. S., Schmid, B., Hutchinson, J. R., & Shefelbine, S. J. (2010). BoneJ: Free and extensible bone image analysis in ImageJ. *Bone*, 47(6), 1076–1079. https://doi.org/10.1016/j. bone.2010.08.023
- Elizabeth, J. S., Broe, K. E., Xu, H., Yang, L., Boyd, S., Biver, E., Szulc, P., Adachi, J., Amin, S., Atkinson, E., Berger, C., Burt, L., Chapurlat, R., Chevalley, T., Ferrari, S., Goltzman, D., Hanley, D. A., Hannan, M. T., Khosla, S., ... Bouxsein, M. L. (2019). Cortical and trabecular bone microarchitecture predicts incident fracture independently of DXA bone minderal density and FRAX in older women and men. *The Bone Microarchitecture International Consortium*, 7(1), 34–43. https://doi.org/10.1016/S2213-8587(18)30308-5.Cortical
- Falk, S., & Dickenson, A. H. (2014). Pain and nociception: Mechanisms of cancer-induced bone pain. *Journal of Clinical Oncology*, 32(16), 1647–1654. https://doi.org/10.1200/JCO.2013.51.7219
- Falk, S., Uldall, M., Appel, C., Ding, M., & Heegaard, A. M. (2013). Influence of sex differences on the progression of cancer-induced bone pain. *Anticancer Research*, 33(5), 1963–1969.
- Faul, F., Erdfelder, E., Lang, A.-G., & Buchner, A. (2007). G\*Power 3: A flexible statistical power analysis program for the social, behavioral, and biomedical sciences. *Behavior Research Methods*, 39(2), 175–191. https://doi.org/10.3758/BF03193146
- Felsenberg, D., & Boonen, S. (2005). The bone quality framework: Determinants of bone strength and their interrelationships, and implications for osteoporosis management. *Clinical Therapeutics*, 27(1), 1–11. https://doi.org/10.1016/j.clinthera.2004.12.020
- Ferguson, V. L., Ayers, R. A., Bateman, T. A., & Simske, S. J. (2003). Bone development and age-related bone loss in male C57BL/6J mice. *Bone*, 33(3), 387–398. https://doi.org/10.1016/S8756-3282(03)00199-6
- Gendron, N., Chahabi, S. Z., Poenou, G., Rivet, N., Belleville-Rolland, T., Lemaire, P., Escuret, A., Ciaudo, M., Curis, E., Gaussem, P., Siguret, V., & Darnige, L. (2019). Pain assessment and factors influencing pain during bone marrow aspiration: A prospective study. *PLoS One*, 14(8), 1–13. https://doi.org/10.1371/journ al.pone.0221534
- Guedon, J. M. G., Longo, G., Majuta, L. A., Thomspon, M. L., Fealk, M. N., & Mantyh, P. W. (2016). Dissociation between the relief of skeletal pain behaviors and skin hypersensitivity in a model of bone cancer pain. *Pain*, 157(6), 1239–1247. https://doi.org/10.1097/j. pain.000000000000514

- Honore, P., & Mantyh, P. W. (2000). Bone cancer pain: From mechanism to model to therapy. *Pain Medicine*, 1(4), 303–309. https://doi.org/10.1046/j.1526-4637.2000.00047.x
- Huang, D., Huang, X., Yan, X., Wu, L., & Wang, M. (2008). Establishment and evaluation of a bone cancer pain model. *Zhong Nan Da Xue Xue Bao. Yi Xue Ban*, 33(2), 115–120.
- Isono, M., Suzuki, T., Hosono, K., Hayashi, I., Sakagami, H., Uematsu, S., Akira, S., Declerck, Y. A., Okamoto, H., & Majima, M. (2011). Microsomal prostaglandin E synthase-1 enhances bone cancer growth and bone cancer-related pain behaviors in mice. *Life Sciences*, 88(15–16), 693–700. https://doi.org/10.1016/j. lfs.2011.02.008
- Ji, H., Pettit, A., Ohmura, K., Ortiz-Lopez, A., Duchatelle, V., Degott, C., Gravallese, E., Mathis, D., & Benoist, C. (2002). Critical roles for interleukin 1 and tumor necrosis factor α in antibody-induced arthritis. *The Journal of Experimental Medicine*, 196(1), 77–85. https://doi.org/10.1084/jem.20020439
- Jimenez-Andrade, J. M., Bloom, A. P., Mantyh, W. G., Koewler, N. J., Freeman, K. T., Delong, D., Ghilardi, J. R., Kuskowski, M. A., & Mantyh, P. W. (2009). Capsaicin-sensitive sensory nerve fibers contribute to the generation and maintenance of skeletal fracture pain. *Neuroscience*, 162(4), 1244–1254. https://doi.org/10.1016/j.neuro science.2009.05.065
- Jones, B. J., & Roberts, D. J. (1968). A rotarod suitable for quantitative measurements of motor incoordination in naive mice. Naunyn-Schmiedebergs Archiv Für Pharmakologie Und Experimentelle Pathologie, 259(2), 211. https://doi.org/10.1007/BF00537801
- Kanaya, K., Iba, K., Abe, Y., Dohke, T., Okazaki, S., Matsumura, T., & Yamashita, T. (2016). Acid-sensing ion channel 3 or P2X2/3 is involved in the pain-like behavior under a high bone turnover state in ovariectomized mice. *Journal of Orthopaedic Research*, 34(4), 566–573. https://doi.org/10.1002/jor.23047
- Karsdal, M. A., Henriksen, K., Sørensen, M. G., Gram, J., Schaller, S., Dziegiel, M. H., Heegaard, A. M., Christophersen, P., Martin, T. J., Christiansen, C., & Bollerslev, J. (2005). Acidification of the osteoclastic resorption compartment provides insight into the coupling of bone formation to bone resorption. *American Journal* of Pathology, 166(2), 467–476. https://doi.org/10.1016/S0002 -9440(10)62269-9
- Kijowski, R., Tuite, M., Kruger, D., Munoz Del Rio, A., Kleerekoper, M., & Binkley, N. (2012). Evaluation of trabecular microarchitecture in nonosteoporotic postmenopausal women with and without fracture. *Journal of Bone and Mineral Research*, 27(7), 1494–1500. https://doi.org/10.1002/jbmr.1595
- Kim, D. H., & Vaccaro, A. R. (2006). Osteoporotic compression fractures of the spine; current options and considerations for treatment. *Spine Journal*, 6(5), 479–487. https://doi.org/10.1016/j. spinee.2006.04.013
- Kim, S., Takahashi, H., Lin, W., Descargues, P., Grivennikov, S., Kim, Y., Luo, J., & Karin, M. (2009). Carcinoma-produced factors activate myeloid cells through TLR2 to stimulate metastasis. *Nature*, 457(7225), 102–106. https://doi.org/10.1038/nature07623
- Kong, Y.-Y., Yoshida, H., Sarosi, I., Tan, H.-L., Timms, E., Capparelli, C., Morony, S., Oliveira-dos-Santos, A. J., Van, G., Itie, A., Khoo, W., Wakeham, A., Dunstan, C. R., Lacey, D. L., Mak, T. W., Boyle, W. J., & Penninger, J. M. (1999). OPGL is a key regulator of osteoclastogenesis, lymphocyte development and lymph-node organogenesis. *Nature*, 397(6717), 315–323. https://doi.org/10.1038/16852
- Kotani, M., Kikuta, J., Klauschen, F., Chino, T., Kobayashi, Y., Yasuda, H., Tamai, K., Miyawaki, A., Kanagawa, O., Tomura, M., & Ishii,

M. (2013). Systemic circulation and bone recruitment of osteoclast precursors tracked by using fluorescent imaging techniques. *The Journal of Immunology*, *190*(2), 605–612. https://doi.org/10.4049/jimmunol.1201345

- Lee, S., Nam, C. M., Yoon, D. H., Kim, K. N., Yi, S., Shin, D. A., & Ha, Y. (2013). Association between low-back pain and lumbar spine bone density: A population-based cross-sectional study Clinical article. *Journal of Neurosurgery: Spine*, 19(3), 307–313. https://doi. org/10.3171/2013.5.SPINE12473
- Li, J., Zeng, L., Xie, J., Yue, Z., Deng, H., Ma, X., Zheng, C., Wu, X., Luo, J., & Liu, M. (2015). Inhibition of osteoclastogenesis and bone resorption in vitro and in vivo by a prenylflavonoid xanthohumol from hops. *Scientific Reports*, 5, 1–14. https://doi.org/10.1038/srep1 7605
- Li, L. H., Wang, Z. C., Yu, J., & Zhang, Y. Q. (2014). Ovariectomy results in variable changes in nociception, mood and depression in adult female rats. *PLoS One*, 9(4), e94312. https://doi.org/10.1371/ journal.pone.0094312
- Li, S., He, H., Ding, M., & He, C. (2010). The correlation of osteoporosis to clinical features: A study of 4382 female cases of a hospital cohort with musculoskeletal symptoms in Southwest China. BMC Musculoskeletal Disorders, 11. https://doi. org/10.1186/1471-2474-11-183
- Lipton, A., & Balakumaran, A. (2012). Denosumab for the treatment of cancer therapy-induced bone loss and prevention of skeletal-related events in patients with solid tumors. *Expert Review of Clinical Pharmacology*, 5(4), 359–371. https://doi.org/10.1586/ecp.12.35
- Mantyh, W. G., Jimenez-Andrade, J. M., Stake, J. I., Bloom, A. P., Kaczmarska, M. J., Taylor, R. N., Freeman, K. T., Ghilardi, J. R., Kuskowski, M. A., & Mantyh, P. W. (2010). Blockade of nerve sprouting and neuroma formation markedly attenuates the development of late stage cancer pain. *Neuroscience*, 171(2), 588–598. https://doi.org/10.1016/j.neuroscience.2010.08.056
- Matsuzaki, K., Udagawa, N., Takahashi, N., Yamaguchi, K., Yasuda, H., Shima, N., Morinaga, T., Toyama, Y., Yabe, Y., Higashio, K., & Suda, T. (1998). Osteoclast differentiation factor (ODF) induces osteoclast-like cell formation in human peripheral blood mononuclear cell cultures. *Biochemical and Biophysical Research Communications*, 246(1), 199–204. https://doi.org/10.1006/ bbrc.1998.8586
- McDonald, M. M., Dulai, S., Godfrey, C., Amanat, N., Sztynda, T., & Little, D. G. (2008). Bolus or weekly zoledronic acid administration does not delay endochondral fracture repair but weekly dosing enhances delays in hard callus remodeling. *Bone*, 43(4), 653–662. https://doi.org/10.1016/j.bone.2008.05.019
- Minett, M. S., Falk, S., Santana-Varela, S., Bogdanov, Y. D., Nassar, M. A., Heegaard, A. M., & Wood, J. N. (2014). Pain without nociceptors? Nav1.7-independent pain mechanisms. *Cell Reports*, 6(2), 301–312. https://doi.org/10.1016/j.celrep.2013.12.033
- Mizuhashi, K., Nagata, M., Matsushita, Y., Ono, W., & Ono, N. (2019). Growth plate borderline chondrocytes behave as transient mesenchymal precursor cells. *Journal of Bone and Mineral Research*, 34(8), 1387–1392. https://doi.org/10.1002/jbmr.3719
- Mundy, G. R. (2002). Metastasis to bone: Causes, consequences and therapeutic opportunities. *Nature Reviews Cancer*, 2(8), 584–593. https://doi.org/10.1038/nrc867
- Newton, P. T., Li, L., Zhou, B., Schweingruber, C., Hovorakova, M., Xie, M., Sun, X., Sandhow, L., Artemov, A. V., Ivashkin, E., Suter, S., Dyachuk, V., El Shahawy, M., Gritli-Linde, A., Bouderlique, T., Petersen, J., Mollbrink, A., Lundeberg, J., Enikolopov, G., Chagin,



A. S. (2019). A radical switch in clonality reveals a stem cell niche in the epiphyseal growth plate. In *Nature* (Vol. 567, Issue 7747, pp. 234–238). Springer, US. doi: 10.1038/s41586-019-0989-6.

- Novack, D. V. (2016). Editorial: Inflammatory osteoclasts: A different breed of bone eaters? *Arthritis & Rheumatology*, 68(12), 2834– 2836. https://doi.org/10.1002/art.39835
- O'Brien, W., Fissel, B. M., Maeda, Y., Yan, J., Ge, X., Gravallese, E. M., Aliprantis, A. O., & Charles, J. F. (2016). RANK-independent osteoclast formation and bone erosion in inflammatory arthritis. *Arthritis* and Rheumatology, 68(12), 2889–2900. https://doi.org/10.1002/ art.39837
- Remeniuk, B., King, T., Sukhtankar, D., Nippert, A., Li, N., Li, F., Cheng, K., Rice, K. C., & Porreca, F. (2018). Disease modifying actions of interleukin-6 blockade in a rat model of bone cancer pain. *Pain*, 159(4), 684–698. https://doi.org/10.1097/j.pain.0000000000 001139
- Sanoja, R., & Cervero, F. (2005). Estrogen-dependent abdominal hyperalgesia induced by ovariectomy in adult mice: A model of functional abdominal pain. *Pain*, 118(1–2), 243–253. https://doi.org/10.1016/j. pain.2005.08.021
- Santy, J., & Mackintosh, C. (2001). A phenomenological study of pain following fractured shaft of femur. *Journal of Clinical Nursing*, 10(4), 521–527. https://doi.org/10.1046/j.1365-2702.2001.00506.x
- Sato, M., Inage, K., Sakuma, Y., Sato, J., Orita, S., Yamauchi, K., Eguchi, Y., Ochiai, N., Kuniyoshi, K., Aoki, Y., Nakamura, J., Miyagi, M., Suzuki, M., Kubota, G., Sainoh, T., Fujimoto, K., Shiga, Y., Abe, K., Kanamoto, H., ... Ohtori, S. (2015). Anti-RANKL antibodies decrease CGRP expression in dorsal root ganglion neurons innervating injured lumbar intervertebral discs in rats. *European Spine Journal*, 24(9), 2017–2022. https://doi.org/10.1007/s00586-015-4058-z
- Savaridas, T., Wallace, R. J., Salter, D. M., & Simpson, A. H. R. W. (2013). Do bisphosphonates inhibit direct fracture healing? *The Bone & Joint Journal*, 95, 1263–1268. https://doi. org/10.1302/0301-620x.95b9.31562
- Schell, H., Lienau, J., Epari, D. R., Seebeck, P., Exner, C., Muchow, S., Bragulla, H., Haas, N. P., & Duda, G. N. (2006). Osteoclastic activity begins early and increases over the course of bone healing. *Bone*, 38(4), 547–554. https://doi.org/10.1016/j.bone.2005.09.018
- Shin, J., Jang, H., Lin, J., & Lee, S. Y. (2014). PKCβ positively regulates RANKL-induced osteoclastogenesis by inactivating GSK-3β.

Molecules and Cells, 37(10), 747–752. https://doi.org/10.14348/ molcells.2014.0220

- Tomimori, Y., Mori, K., Koide, M., Nakamichi, Y., Ninomiya, T., Udagawa, N., & Yasuda, H. (2009). Evaluation of pharmaceuticals with a novel 50-hour animal model of bone loss. *Journal of Bone* and Mineral Research, 24(7), 1194–1205. https://doi.org/10.1359/ jbmr.090217
- Topoliński, T., Mazurkiewicz, A., Jung, S., Cichański, A., & Nowicki, K. (2012). Microarchitecture parameters describe bone structure and its strength better than BMD. *The Scientific World Journal*, 2012, 1–7. https://doi.org/10.1100/2012/502781
- Yang, H., Yan, H., Li, X., Liu, J., Cao, S., Huang, B., Huang, D., & Wu, L. (2018). Inhibition of connexin 43 and phosphorylated NR2B in spinal astrocytes attenuates bone cancer pain in mice. *Frontiers in Cellular Neuroscience*, *12*, 1–14. https://doi.org/10.3389/ fncel.2018.00129
- Zhou, S., Wang, G., Qiao, L., Ge, Q., Chen, D., Xu, Z., Shi, D., Dai, J., Qin, J., Teng, H., & Jiang, Q. (2018). Age-dependent variations of cancellous bone in response to ovariectomy in C57BL/6J mice. *Experimental and Therapeutic Medicine*, 15(4), 3623–3632. https:// doi.org/10.3892/etm.2018.5839
- Zilber, S., Epstein, N. J., Lee, S. W., Larsen, M., Ma, T., Smith, R. L., Biswal, S., & Goodman, S. B. (2008). Mouse femoral intramedullary injection model: Technique and microCT scan validation. *Journal* of Biomedical Materials Research. Part B, Applied Biomaterials, 84(1), 286–290. https://doi.org/10.1002/jbm.b.30872

#### SUPPORTING INFORMATION

Additional supporting information may be found online in the Supporting Information section.

How to cite this article: de Clauser L, Santana-Varela S, Wood JN, Sikandar S. Physiologic osteoclasts are not sufficient to induce skeletal pain in mice. *Eur. J. Pain.* 2020;00:1–14. <u>https://doi.</u> org/10.1002/ejp.1662